Author’s response to reviews

Title: Causes of higher symptomatic airway load in patients with Chronic Rhinosinusitis

Authors:
Øystein Eskeland (oystein.eskeland@gmail.com)
Kjell-Arild Danielsen (k.a.danielse@medisin.uio.no)
Fredrik Dahl (fdah@ahus.no)
Katrin Fridrich (Katrin.fridrich@ahus.no)
Vivian Orszagh (vivian.cecilie.orszagh@ahus.no)
Gregor Bachmann-Harildstad (gregor.bachmann-harildstad@medisin.uio.no)
Espen Burum-Auensen (espen.burum-auensen@medisin.uio.no)

Version: 4 Date: 13 Nov 2017

Author’s response to reviews:

Dear Editor

Thank you for your comments and your interest in our manuscript.

The manuscript has been revised according to your comments.

1) In methods... you write 61 patients were diagnosed with CRS... 27 of..35 of.. which is correct; 61? 62?

The right number is of course 62 and has been changed accordingly.

(page 6, line6)

2) "Materials and methods " should be amended with "Methods".

According to the editor recommendations we have changed the subheading to Methods.
We hope that you will accept the revisions done, and consider the new manuscript for publication.

Sincerely Yours

Øystein Eskeland

MD, Otorhinolaryngology